Author's response to reviews

Title: The third Symptom Management Research Trial in Oncology (SMaRT Oncology-3): a randomised trial to determine the efficacy of adding a complex intervention for major depressive disorder (Depression Care for People with Lung Cancer) to usual care, compared to usual care alone in patients with lung cancer.

Authors:

Jane Walker (jane.walker@ed.ac.uk)
Michael Sharpe (michael.sharpe@ed.ac.uk)
Jim Cassidy (j.cassidy@beatson.gla.ac.uk)

Version: 2 Date: 22 July 2009

Author's response to reviews:

Thank you for your review. The authors are unable to provide a full detailed analysis plan at this stage as this will will be informed by the ongoing analysis of the SMaRT Oncology-3 Pilot Trial. However, we have endeavoured to address the reviewer's specific comments in the revised manuscript attached.